Author's response to reviews

Title: Impact of a web-assisted tobacco quality improvement intervention of subsequent patient tobacco product use: A National Dental PBRN Study

Authors:

Thomas K Houston II (thomas.houston@umassmed.edu)
Kathryn L DeLaughter (kathryn.delaughter@umassmed.edu)
Midge N. Ray (midgeray@uab.edu)
Gregg H. Gilbert (ghg@umassmed.edu)
Jeroan J. Allison (jeroan.allison@umassmed.edu)
Catarina I. Kiefe (catarina.kiefe@umassmed.edu)
Julie E. Volkman (julie.volkman@va.gov)

Version: 4 Date: 8 October 2012

Author's response to reviews:

Please find attached a revised manuscript. The only change made is the addition of a very recent new network name.